Author's response to reviews

Title: Explanations of socioeconomic differences in changes in physical function in older adults: results from the Longitudinal Aging Study Amsterdam

Authors:

Annemarie Koster (a.koster@zw.unimaas.nl)
Hans Bosma (hans.bosma@zw.unimaas.nl)
Marjolein I. Broese van Groenou (mi.broese@fsw.vu.nl)
Gertrudis I.J.M. Kempen (g.kempen@zw.unimaas.nl)
Brenda W.J.H. Penninx (Brendap@ggzva.nl)
Jacques TH.M. van Eijk (j.vaneijk@zw.unimaas.nl)
Dorly J.H. Deeg (djh.deeg@vumc.nl)

Version: 2  Date: 1 December 2005

Author's response to reviews:

Dear Dr. Phillips,

As requested in your e-mail of November 17 we clarified what form of consent was obtained and that ethical approval was obtained for the study. We added this to the study sample section of the methods (page 5)

We also removed the initials within the authors list in the author detail section.

Yours sincerely,

Annemarie Koster